Author's response to reviews

Title: Evaluating a facilitation model to promote organisational development in primary care: The Maturity Matrix.

Authors:

Melody Rhydderch (RhydderchM@Cardiff.ac.uk)
Adrian Edwards (edwardsag@Cardiff.ac.uk)
Martin Marshall (martin.marshall@manchester.ac.uk)
Glyn Elwyn (elwynG@cardiff.ac.uk)
Grol Richard (r.grol@wok.umcn.nl)

Version: 2 Date: 12 January 2006

Author's response to reviews:

Melody Rhydderch

Occupational Psychologist

NHS primary care research training fellow

Centre for Health Sciences Research

Cardiff University

56 Park Place

Cardiff

CF10 3AT

12th January, 2005

Alex J Pemberton PhD

Assistant Editor

BMC-series Journals

BioMed Central Ltd

Middlesex House

34-42 Cleveland Street

London, W1T4LB

Dear Alex

Revision to manuscript

MS: 6420869818745274: Evaluating a facilitation model to promote organisational development in primary care: The Maturity Matrix.
Thank-you for your email dated 9th January regarding ethical approval for the above research. The study forms part of a series of studies into the validity of the Maturity Matrix conducted as part of my PhD thesis. Ethical approval was sought and granted from an MREC in 2002. I have amended the text in the methods section accordingly, page 5, paragraph 1.

I look forward to hearing from you.

Yours sincerely

Melody Rhydderch